Author's response to reviews

Title: Clinical presentation of abdominal tuberculosis in HIV seronegative adults

Authors:

Cengiz Bolukbas (drcengizbolukbas@hotmail.com)
Fusun F Bolukbas (fusunbol@yahoo.ca)
Tulin Kendir (eylul19@hotmail.com)
Remzi A Dalay (remzidalay@att.net)
Nihat Akbayir (akbayirn@hotmail.com)
Mehmet H Sokmen (hmsokmen@hotmail.com)
Ali T Ince (alince@superonline.com)
Mithat Guran (mithatguran@yahoo.com)
Erkan Ceylan (drrerkanceylan@yahoo.com)
Guray Kilic (gurayk61@hotmail.com)
Oya Ovunc (hnhgastro@superonline.com)

Version: 3  Date: 18 June 2005

Dear Editor,

Thanks for your attention that you have paid to our manuscript (MS: 1275370683621661) entitled as "Clinical presentation of abdominal tuberculosis in HIV seronegative adults". Please find attached manuscript that was revised as formatting changes requested. The changes were provided with a point-by-point description at below:

oyaovunc@ekolay.net - this author email was updated in the submission system.
Author list - The list of authors in the manuscript has been written exactly as they are in the submission system.
Affiliations - Each affiliation has been written in full and been provided as requested.
Abstract - The space between others and demographic in the methods subsection of the abstract in the submission system and the manuscript file was corrected.
Section headings - We removed all unnecessary capitals in section headings.
Highlighting/Tracking - We removed any highlighting/tracking from our manuscript text.

Checking for typing error- We checked our manuscript carefully for any typing errors.

Yours Sincerely,

Corresponding Author:
Cengiz Bolukbas

Yenisehir mah. 4. sok. Altiparmak apt. Kat.2 Daire 3
63100 Yenisehir/ Sanliurfa TURKEY
Phone:+90 414 3155250
Fax: +90 414 3151181
e-mail: drcengizbolukbas@hotmail.com